Author's response to reviews

Title: Primary paediatric care models and non-urgent Emergency Department utilization: an area-based cohort study

Authors:

Sara Farchi (farchi@asplazio.it)
Arianna Polo (polo@asplazio.it)
Francesco Franco (franco@asplazio.it)
Domenico Di Lallo (dilallo@asplazio.it)
Gabriella Guasticchi (guasticchi@asplazio.it)

Version: 7 Date: 19 April 2010

Author's response to reviews: see over
Rome, 19 April 2010

Dear Editor,

We would like to submit the revised version of the manuscript entitled “Primary Pediatric Care models and non-urgent Emergency Department utilization: An area-based cohort study,” for your consideration for publication in BMC Family Practice.

According to the journal style we formatted the paper.

On behalf of the authors

Sara Farchi